Author's response to reviews

Title: Prevalence and risk factors of Post Traumatic Stress Disorder seven years after the conflict in three districts in northern Uganda (The Wayo-Nero Study)

Authors:

  James Mugisha (jmmugi77@hotmail.com)
  Herbert Muyinda (muyindaherbert@hotmail.com)
  Peter Wandiembe (symon.wand@gmail.com)
  Eugene Kinyanda (Eugene.Kinyanda@mrcuganda.org)

Version: 2 Date: 8 January 2015

Author's response to reviews:

Tables have now been put at the end of the manuscript and a cover letter provided